Author’s response to reviews

Title: Diagnostic efficacy of double-balloon enteroscopy in patients with suspected isolated small bowel Crohn’s disease

Authors:
Zihan Huang (hzhan730630@163.com)
Xiang Liu (liux@sj-hospital.org; liux_2017@163.com)
Fei Yang (lakeblue18@163.com)
Guoxin Wang (wanggx@sj-hospital.org)
Nan Ge (gen@sj-hospital.org)
Sheng Wang (wangs@sj-hospital.org)
Jintao Guo (guojt@sj-hospital.org)
Siyu Sun (sunsy@sj-hospital.org)

Version: 5 Date: 03 Feb 2020

Author’s response to reviews:

Dear Editor:

Thank you very much for your help in processing the review of our manuscript (Manuscript ID: BMGE-D-19-00493R4). We have carefully read your comments and modified the manuscript based on your comments. The itemized responses to the comments are listed below. We hope that all these corrections and revisions will be satisfactory. Once again, thank you for reviewing this article and for your valuable suggestions.

1. Title: Diagnostic efficacy of double-balloon enteroscopy in patients with suspected isolated small bowel Crohn’s disease

2. Manuscript type: Article

3. Corresponding author: Xiang Liu

4. Full author names: Zihan Huang, Xiang Liu, Fei Yang, Guoxin Wang, Nan Ge, Sheng Wang, Jintao Guo, Siyu Sun

Sincerely,
Xiang Liu  
Endoscopy Center  
China medical University,  
Shenyang, Liaoning, 110000, P R China.

Responses to comments  
Point by point response to editor:

1. Please add a “Conclusions” section after the “Discussion” section. This should state clearly the main conclusions of the research article and give a clear explanation of their importance and relevance. 
Response:  
Thank you for your thorough revisions.  
As you pointed out, we have added a “Conclusions” section after “Discussion” section.

2. Please note, the "Availability of Data and Materials" declaration refers to the raw data used in your study. The statement currently provided is not correct in the case of your manuscript. We strongly encourage all authors to share their raw data, either by providing it in a supplementary file or depositing it in a public repository and providing the details on how to access it in this section. If you do not wish to share your data, please clearly state this in this section along with a justification. For more information and a list of suitable availability statements, please see our Submission guidelines: https://www.biomedcentral.com/getpublished/editorial-policies#availability+of+data+and+materials  
Response:  
Thank you for your comment.  
We have modified the statement in “Availability of data and materials” section.

3. Please remove the ages and gender from table 1 to help maintain participant confidentiality. Please use age ranges but not gender in their place. Please check that the identifiers do not link to patients.  
Response:  
Thank you for your comment.  
We have removed the ages and gender and used age ranges in Table 1. We have checked that the identifiers do not link to patients.

4. At this stage, please upload your manuscript as a single, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.  
Response:  
Once again, thank you for your comments.  
We have modified the manuscript and table to ensure that the uploaded file is the final clean version.